Author's response to reviews

Title: Genetic polymorphisms in MDR1, CYP3A4 and CYP3A5 genes in a Ghanaian population: a plausible explanation for altered metabolism of ivermectin in humans?

Authors:

William Kudzi (wkudzi@yahoo.com)
Alexander N.O Dodoo (alexooo@yahoo.com)
Jeremy J Mills (Jeremy.mills@port.ac.uk)

Version: 6 Date: 13 July 2010

Author's response to reviews:

Dear Editor,

I would like to resubmit the revised manuscript MS: 2123107729263686 titled "Genetic polymorphisms in MDR1, CYP3A4 and CYP3A5 genes in a Ghanaian population: a plausible explanation for altered metabolism of ivermectin in humans?"

I have revised the section on acknowledgement to read "The efforts of Dr Kwabla Ahadzi of OCRC in Hohoe and Samuel Ahorhorlu of Centre for Tropical Clinical Pharmacology and Therapeutics in collecting samples are greatly appreciated. This work was supported by the Government of Ghana through a GETFUND Scholarship Award"

Thank you